Author’s response to reviews

Title: Anti-hyperalgesic properties of a flavanone derivative Poncirin in acute and chronic inflammatory pain models in mice

Authors:

Ruqayya Afridi (r.afridi36@gmail.com)
Ashrafullah Khan (ashrafwazir6@gmail.com)
Sidra Khalid (sidra.merlin@gmail.com)
Busha Shal (bushra.shal@gmail.com)
Hina Rasheed (hinarasheed137@gmail.com)
Zia Ullah (ziaullah121@gmail.com)
Omer Shehzad (omersnu@yahoo.com)
Yeong Shik Kim (kims@snu.ac.kr)
Salman Khan (skhan@qau.edu.pk)

Version: 7 Date: 22 Jul 2019

Author’s response to reviews:

July 22, 2019

BMC-Pharmacology and Toxicology

Editorial Office

RE: Revision PHAT-D-18-00110R6

Dear Editor,

We highly appreciate for providing the valuable comments which help us to improve our manuscript greatly. For publication in BMC-Pharmacology and Toxicology. We have addressed the entire shortcoming in our study in earnest, in the revised manuscript. All the changed/corrected parts have been highlighted in the text of the revised manuscript.

All the points raised by the editor are appended below.
Editor Comments:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in the Methods section. While we understand that you may wish to express some of the same ideas contained in this publication, please be aware that we cannot condone the use of text from previously published work. If there is overlap in the Methods section, please ensure to summarize the methods and cite the source. Please rephrase these sections to minimise overlap.

Response:

Editor is highly appreciated for the valuable suggestion. The materials and methods section was modified and rephrased as per recommendation in the revised manuscript. The author tried level best to avoid any texture overlap with the previous published data

"N-Pyrazoloyl and N-thiopheneacetyl hydrazone of isatin exhibited potent anti-inflammatory and anti-nociceptive properties through suppression of NF-κB, MAPK and oxidative stress signaling in animal models of inflammation." Inflammation Research 68.7 (2019): 613-632.

The methods section were summarized and cited by appropriate reference as per recommendation.

2. Thank you for providing a response to the reviewers. As this document is no longer required at this stage of the publication process, please remove your ‘Authors response letter’ from your submission’s supplementary files.

Response:

The author response letters has been removed as per recommendation.

3. We note that two authors have the same initials (SK). To distinguish between them in the Authors’ Contributions section please designate them as SK1 and SK2, with SK1 corresponding to Sidra Khalid, chosen because this name is furthest up the author list.

Response:
The author contribution section the initials were changed as per recommendation in the revised manuscript. The Sidra Khalid was designated as SK1 and Salman Khan was designated as SK2.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

Response:

The manuscript was uploaded as single, final, clean and the changes will not be highlighted. All the tables and figures was uploaded as clean version and uploaded as separate file. All the figures and tables are appropriately cited within the text as per recommendation in the revised manuscript.

We are grateful for your efforts for the improvement of our manuscript and are hopeful for a positive decision this time, on our manuscript.

The final version of the manuscript was edited by the native speaker (www.aje.com).

We would be pleased to consider our manuscript for publication in BMC-Pharmacology and Toxicology.

Sincerely yours

Salman Khan (PharmD, Ph.D.)

Assistant Professor of Pharmacology

Department of Pharmacy, Faculty of Biological Sciences, Quaid-i-Azam University, Islamabad, Pakistan

Address: Room No. 03, Department of Pharmacy, Quaid-i-Azam University, Islamabad, Pakistan

Phone: +92-51-90644207

Email: skhan@qau.edu.pk